Author's response to reviews

Title: Etanercept in the treatment of recalcitrant enteropathic arthritis : a case report

Authors:

Mohd Shahrir Mohamed Said (drobiwan@gmail.com)
Sazliyana Shaharir (sazliyana@hotmail.com)
Sakthiswary Rajalingham (sakthis5@hotmail.com)
Sheikh Anwar Abdullah (sheikhanwar@gmail.com)
Ngii Chai Soon (csngiu@gmail.com)
Aizan Hassanudin (aizan@gmail.com)
Mohd Shahdan Shahid (mohdshahdan@gmail.com)

Version: 8 Date: 3 August 2011

Author's response to reviews: see over
Case report
Etanercept in the treatment of recalcitrant enteropathic arthritis : a case report
Mohd Shahrir Mohamed Said, Sazliyana Shaharir, Sakthiswary Rajalingham, Sheikh Anwar Abdullah, Ngiu Chai Soon, Aizan Hassanudin and Mohd Shahdan Shahid
The Deputy Editor who is handling your manuscript would like to pass on the following comments:
"I have read this revised manuscript. The authors have made most of the recommended revisions with the exception of the title which needs to be changed and a few others. Also, there were some more minor grammatical errors that will need correcting that I found upon reading the paper again:
1. Please change the title to the following: "Etanercept in the treatment of recalcitrant enteropathic arthritis : a case report."
2. In the introduction section, the sentence "For peripheral arthritis, two types has been described in literature, oligoarticular (type I) and polyarticular (type II) (7-9)." please place the word "the" before "literature".
3. In the introduction section the following sentence: "Type I usually involve large joints of lower limbs and is associated with IBD flares, meanwhile type II arthritis, is polyarticular and symmetric, usually evolves in chronic disease, and involves not only mainly hands and feet but also large joints." should be: "Type I usually involves large joints of lower limbs and is associated with IBD flares, meanwhile type II arthritis, is polyarticular and symmetric, usually evolves in chronic disease, and involves not only mainly hands and feet but also large joints."
4. Please have the authors use square brackets when citing references within the text as per JMCR style.
5. The term "et al" should be spelled "et al." The authors have done this only incompletely: For example, in the discussion section the following sentences: "Amongst psoriatic arthritis patients Bravo et al demonstrated that at a 10-year time horizon, the etanercept was
more cost effective than infliximab." should be:

"Amongst psoriatic arthritis patients, Bravo et al. demonstrated that at a 10-year time horizon, the etanercept was more cost effective than infliximab."

AND:

"A study by Baraliakos et al. showed that after 2 years of treatment, MRI evaluation showed minor spinal inflammation was still present in 64% of the patients [29]."

6. In the discussion section for the sentence: "Etanercept may improve joint symptoms but studies don't show any benefit for bowel inflammation (16)" there should be a period at the end of the sentence.

7. In the caption for Figure 1: "Fragment of fibrocollagenous connective tissue stroma admixed with fibrinous exudates and abundant foci of vascular proliferation indicates inflammatory process." should be:

"Fragment of fibrocollagenous connective tissue stroma admixed with fibrinous exudates and abundant foci of vascular proliferation indicates an inflammatory process."

8. Lastly, please have the authors have the manuscript read by a native English speaker to further discover (if any) minor grammatical errors that I may have missed."

The Editorial Team would also like to request for formatting changes required your manuscript. Kindly address the following:

1. Please replace dates found in your manuscript to the amount of time this occurred before the presentation of the case report.

2. Please remove figure 1 found before reference list. Figure uploaded as a separated file is enough. Please could you modify the manuscript and submit a revised version by logging on to the website, http://www.biomedcentral.com/my/manuscripts. We encourage you do this as soon as possible to minimise the delay to the review of your manuscript. Your manuscript will not be peer reviewed until the revised manuscript has been received. We look forward to hearing from you by August 9, 2011. If you have any queries please don’t hesitate to get in
Dear Editor and staff,

I have corrected all that had been asked and also rectify the English as well.

Thank you.